**RESEARCH ARTICLE** 

# Estimating the burden of scrub typhus: A systematic review

## Ana Bonell<sup>1</sup>, Yoel Lubell<sup>2,3</sup>, Paul N. Newton<sup>3,4</sup>, John A. Crump<sup>5</sup>, Daniel H. Paris<sup>2,3,6,7</sup>\*

 Oxford University Clinical Research Unit, National Hospital of Tropical Diseases, Hanoi, Vietnam,
Mahidol Oxford Tropical Medicine Research Unit, Mahidol University, Bangkok, Thailand, 3 Centre for Tropical Medicine and Global Health, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 4 Lao-Oxford-Mahosot Hospital-Wellcome Trust Research Unit, Microbiology Laboratory, Mahosot Hospital, Vientiane, Laos, 5 Centre for International Health, University of Otago, Dunedin, New Zealand,
Department of Medicine, Swiss Tropical and Public Health Institute, Basel, Switzerland, 7 Faculty of Medicine, University Basel, Basel, Switzerland

\* daniel.paris@swisstph.ch

# Abstract

## Background

Scrub typhus is a vector-borne zoonotic disease that can be life-threatening. There are no licensed vaccines, or vector control efforts in place. Despite increasing awareness in endemic regions, the public health burden and global distribution of scrub typhus remains poorly known.

## Methods

We systematically reviewed all literature from public health records, fever studies and reports available on the Ovid MEDLINE, Embase Classic + Embase and EconLit databases, to estimate the burden of scrub typhus since the year 2000.

## Findings

In prospective fever studies from Asia, scrub typhus is a leading cause of treatable nonmalarial febrile illness. Sero-epidemiological data also suggest that *Orientia tsutsugamushi* infection is common across Asia, with seroprevalence ranging from 9.3%–27.9% (median 22.2% IQR 18.6–25.7). A substantial apparent rise in minimum disease incidence (median 4.6/100,000/10 years, highest in China with 11.2/100,000/10 years) was reported through passive national surveillance systems in South Korea, Japan, China, and Thailand. Case fatality risks from areas of reduced drug-susceptibility are reported at 12.2% and 13.6% for South India and northern Thailand, respectively. Mortality reports vary widely around a median mortality of 6.0% for untreated and 1.4% for treated scrub typhus. Limited evidence suggests high mortality in complicated scrub typhus with CNS involvement (13.6% mortality), multi-organ dysfunction (24.1%) and high pregnancy miscarriage rates with poor neonatal outcomes.



# 

**Citation:** Bonell A, Lubell Y, Newton PN, Crump JA, Paris DH (2017) Estimating the burden of scrub typhus: A systematic review. PLoS Negl Trop Dis 11(9): e0005838. https://doi.org/10.1371/journal. pntd.0005838

**Editor:** Janet Foley, University of California Davis, UNITED STATES

Received: March 13, 2017

Accepted: July 28, 2017

Published: September 25, 2017

**Copyright:** © 2017 Bonell et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information file.

**Funding:** Mahidol Oxford Tropical Medicine Research Unit (MORU), Lao-Oxford-Mahosot Hospital-Wellcome Trust Research Unit (LOMWRU) and Oxford University Clinical Research Unit Vietnam (OUCRU) are in part supported by the Wellcome Trust, UK. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

### Interpretation

Scrub typhus appears to be a truly neglected tropical disease mainly affecting rural populations, but increasingly also metropolitan areas. Rising minimum incidence rates have been reported over the past 8–10 years from countries with an established surveillance system. A wider distribution of scrub typhus beyond Asia is likely, based on reports from South America and Africa. Unfortunately, the quality and quantity of the available data on scrub typhus epidemiology is currently too limited for any economical, mathematical modeling or mapping approaches.

## Author summary

Scrub typhus is a mite-transmitted infectious disease that can be life-threatening. Diagnosing this disease is difficult, requiring special techniques that are often not readily available. As the actual impact of scrub typhus on the population and its geographical distribution remains unknown, we searched systematically for available information in medical databases. Scrub typhus is common: more than every fifth person in areas where scrub typhus occurs carry antibodies as a sign of previous contact. All countries with an established surveillance system have recorded an increase in scrub typhus cases over the past 8–10 years, while reports from South America and Africa suggest a wider distribution beyond Asia. Scrub typhus is a serious disease: approximately 6% of cases die if untreated, and 1.5% if treated, but mortality can reach 13% in areas where the usual treatment does not always work well. Death rates of complications are higher, reaching 14% in brain infections, 24% with multiple organ failure, and pregnancies with scrub typhus can have poor outcomes, with high miscarriage rates. Despite many limitations on the amount and quality of available reports, we found that scrub typhus is a severely underappreciated tropical disease, affecting mainly rural populations, but increasingly urban areas as well.

# Introduction

Scrub typhus is an infectious disease caused by *Orientia tsutsugamushi*, an obligate intracellular bacteria, transmitted by the bites of chigger mites [1]. In Southeast Asia, scrub typhus is a leading cause of treatable non-malarial febrile illness [2]. The first accounts linking febrile illness with the appearance of "harmful" mites (Japanese: "tsutsuga" mushi) range back to 313 AD in China [3]. Scrub typhus was originally associated with the Asian-Pacific "Tsutsugamushi triangle," until recent evidence from the Arabian Peninsula, Chile and possibly Kenya suggested a wider global distribution in tropical and subtropical regions [4–7].

The use of improved diagnostic methods, increased medical investigations and awareness have recently contributed to greater recognition of scrub typhus in some countries, such as in Laos, India, southern China, South Korea, and Japan [8]. There is also evidence suggesting that a combination of climate change and expansion of humans into previously uninhabited areas may play a role in both re-emergence and apparent rising incidence of scrub typhus [9–11].

There are no licensed vaccines for scrub typhus, and no systematic vector control efforts in place. Despite increasing awareness in endemic regions, the public health burden and global distribution of scrub typhus remains poorly known.

Although scrub typhus received much attention before and during the Second World War and to a lesser degree during the Vietnam/American war, basic epidemiology is poorly understood with limited data on incidence and burden of disease for patients, their families, societies and the economy. This ignorance is probably due to a combination of factors; clinical presentation is very similar to other causes of fever, diagnostic difficulties contribute to mis-diagnosis and under recognition, and appropriate diagnostic tests are not widely available. Following the discovery of chloramphenicol in the 1940s, the scientific interest dropped rapidly and scrub typhus has since received little global attention [12]. The data quoted by the World Health Organization (WHO) stating that over a billion people are at risk and one million cases are estimated per year is referenced to a paper published 20 years ago in 1997 [13, 14].

Extrapolation based on geographical mite distributions and densities are not helpful due to patchy data, limited by the dynamics of infected mite populations and insufficient characterization of transmitting vectors. With new data and improvements in approaches to estimating the burden of febrile illnesses, it is important to reevaluate the burden of scrub typhus.

Rationale for this study: Scrub typhus is among the leading causes of undifferentiated treatable fever in Asia. The mortality rates appear low at first glance, but considering the numbers of those exposed and/or infected a significant disease burden is expected globally. The following research questions were addressed: What is the estimated global burden of disease for scrub typhus? What data on seroprevalence and minimum incidence for scrub typhus are available by geographical regions? What data on DALYs, YLLs and YLDs are available, and what is the mortality rate of treated scrub typhus?

In this study we summarized the literature relating to the disease burden and economic impact of scrub typhus since the year 2000 in order to estimate the global incidence and burden of this disease.

## Methods

A literature search of three databases: Ovid MEDLINE (2000-present), Embase Classic + Embase (2000-present) and EconLit (2000-present) was conducted on 11<sup>th</sup> April 2016 using three search strategies. First search terms: Scrub typhus, *Orientia tsutsugamushi, Rickettsia tsutsugamushi*, chigger borne rickettsiosis, chigger borne typhus, *Orientia tsutsugamushi* infection, *Rickettsia tsutsugamushi* infection, tsutsugamushi disease, tsutsugamushi fever (keyword) AND prevalence, incidence, epidemiology. A second search included the above search for scrub typhus and all variations AND cost, cost analysis, cost of illness, drug costs, economics, health care cost, hospital costs, cost benefit analysis, cost effectiveness analysis, quality adjusted life year. A third search included scrub typhus AND mortality or death on the 1<sup>st</sup> Oct 2016, for which all currently available data was included (no date restrictions). Data on untreated mortality have been reported [15], and therefore only papers with treated infection were included in estimating mortality. All titles and abstracts were reviewed by 2 authors for inclusion and any disagreements were discussed and inclusion based on the senior author's opinion. Only English language publications were included.

A total of 190 publications were selected for full article review (Fig 1A and 1B, S1 File). The final number of articles included for full data extraction was 87. The data extraction form was trialed on the first 5 papers and required minor alterations. Due to the limited nature of data available no summary measures were applied. Studies were examined for selection bias and graded as follows:

1. Population based active surveillance / community surveillance / hospital surveillance.



A PRISMA Flow Diagram - Search for scrub typhus epidemiology and health economics from 2000 onwards.

#### B PRISMA Flow Diagram - Search for scrub typhus mortality with no date restrictions.

Fig 1. PRISM flow diagrams summarizing the search results. Panel A, search for scrub typhus epidemiology and health economics from 2000 onwards. Panel B, search for scrub typhus mortality with no date restrictions. Flow diagrams were downloaded from <a href="https://www.prisma-statement.org">www.prisma-statement.org</a>.

https://doi.org/10.1371/journal.pntd.0005838.g001

- 2. Prospective, consecutive patient case series with no inappropriate exclusions / retrospective case series / non-consecutive case series / reference laboratory series.
- 3. Exclusion of patients (i.e. most unwell patients treated) likely to significantly affect outcome / summary of case reports from literature.

Papers were also graded on diagnostic tests used:

- 1. Pathogen detection through cell culture/animal inoculation. PCR positivity. IFA serological diagnosis with either sero-conversion or fourfold antibody response.
- 2. Single high Weil Felix titre of ≥1:100 for all samples / Single high IFA titre >1:50 for all samples / IgM or IgG ELISA positive.
- 3. Single Weil Felix or IFA titre, non-significant for all patients / No record of significant Weil Felix or IFA titre / Clinical diagnosis / unspecified.

## Results

Of the 87 studies included, 44 (50.6%) gave information on incidence, seroprevalence and/or prevalence in febrile inpatients (denominator = febrile cases per year), whilst health economic or burden of disease data were given in 4 studies (4.6%) and mortality data in 38 (43.7%). The publications with no apparent patient selection bias and use of grade A evidence to diagnose scrub typhus were few (16/87, 18.4%). A total of 143,544 patients with scrub typhus were described in the included papers. Females were reported to be more commonly infected than males 77,204 versus 57,535 (57.3% versus 42.7%, respectively).

## Incidence surveillance data

Five countries report a passive national surveillance system for scrub typhus.

In South Korea scrub typhus was designated a group III notifiable disease (requiring mandatory reporting and routine monitoring) in 1994. Cases are confirmed by the Korean Centre for Disease Control and Prevention (KCDC) and must show one of the following: an increase in the IFA IgM to *O. tsutsugamushi* of  $\geq$  1:16; an increase in the anti-*O. tsutsugamushi* IFA IgG titre to  $\geq$ 1:256; a  $\geq$  4 fold increase in IFA titre. Data from KCDC suggest that the annual minimum incidence increased from 5.7 to 17.7/100,000 people from 2001 to 2012 (>3-fold) (Table 1) [16–18]. Interestingly, the number of patients recorded in urban areas has also increased dramatically, for example, the annual minimum incidence in Ulsan Metropolitan City increased from 2.8/100,000 in 2003 to 59.7/100,000 in 2013 (>21 fold). In Seoul there is evidence of urban scrub typhus, further demonstrating the changing geographical scope and habitat of infected chigger mites [16].

<u>In Japan</u> scrub typhus is a notifiable disease and must be reported to the National Epidemiological Surveillance of Infectious Diseases (NESID) within 7 days of diagnosis by a physician. Confirmed cases are based on: isolation or identification of the organism in the blood; PCR positivity; detection of serum IgM;  $a \ge 4$  fold increase in IFA titre. Data from NESID show an increase of annual minimum incidence from 0.6/100,000 in 2000 to 3.6/100,000 in 2008 (6-fold) [19, 20].

<u>In Thailand</u>, scrub typhus patients have been reported to the Bureau of Epidemiology for the last 30 years. Data can be viewed online on the homepage available under URL: <u>http://</u>www.boe.moph.go.th/boedb/surdata/disease.php?dcontent=situation&ds=44

Cases are defined based on one or more of the following: isolation or identification of the organism in the blood or tissue sample; PCR positivity;  $a \ge 4$  fold increase in IFA titre (IgG and/or IgM);  $\ge 1:400$  IFA in acute serum (IgG and/or IgM); IgM ELISA positivity. Data from the Bureau of Epidemiology noted an increase of annual minimum incidence from 6.0/100,000 in 2003 to 17.1/100,000 in 2013 (2.9 fold) [21].

In China, scrub typhus is a notifiable disease that must be reported to the China Center for Disease Control and Prevention. Cases are defined as those with clinically compatible infection and one or more of the following; isolation or identification of the organism in a blood or tissue sample; PCR positivity;  $a \ge 1:160$  Weil-Felix test;  $a \ge 4$  fold increase in IFA titre (IgG and/ or IgM). The reported countrywide minimum incidence increased from 0.1/100,000 to 1.1/ 100,000 people/year from 2006 to 2014 (>11-fold) [22]. The reported incidence rates vary widely by region with the southern provinces more affected. Guangdong Province saw an increase in reported annual minimum incidence from 0.4/100,000 to 3.6/100,000 people from 2006 to 2013 (>8-fold), whereas in 2012 the provinces of Laiwu and Guangzhou City had annual incidences of 5.5/100,000 and 9.9/100,000 people, respectively [10, 23–25].

## Seroprevalence data

There are seroprevalence data available from Bangladesh, Indonesia, Laos, Malaysia, Papua New Guinea and Sri Lanka (Table 1). Seropositivity ranged from 9.3%-27.9% suggesting high background exposure levels to *O. tsutsugamushi* in these countries [26–31]

## **Fever studies**

There are several case series describing the frequency of scrub typhus among patients presenting with fever. In India, scrub typhus was the causative agent in 16.1–96.9% of febrile patients presenting to hospitals (<u>Table 2</u>). However, these studies all suffer from selection bias, as other causes of febrile illness had already been excluded. Studies from Cambodia, Laos, Nepal, and

First Author	Country	Incidence/ 100,000	Year data collected	Total number infected	Patient Selection Grade	Diagnostic Grade
Park	South Korea	17.7	2012	8,604	1	A
Yasunaga	Japan	3.6	2007–2008	210	3	A
Wu	China	1.22	2014	16,050	1	A
Lee	Taiwan	14.3	2000–2004	1,396	1	A
N/A	Thailand	11.8	2015	7,696	1	A
First Author	Country, Region	Sero- prevalence	Year data collected	Total number tested	Patient Selection Grade	Seroprevalence Diagnostic Grade
Maude	Bangladesh	23.7%	2010	1,209	1	В
Richards	Indonesia, Gag Island	9.3%	2003	53	2	В
Vallée	Laos, Vientiane	20.7%	2006	2,002	1	В
Тау	Malaysia, Western Malaysia	17.9%	2007–2010	280	2	В
Spicer	Papua New Guinea	27.9%	2001	140	2	В
Premaratna	Sri Lanka	26.3%	2008	57	3	В

#### Table 1. Estimates of incidence and sero-prevalence per country.

Patient selection grade: Grade I = Population based active surveillance / community surveillance / hospital surveillance, Grade II = Prospective, consecutive patient case series with no inappropriate exclusions / non-consecutive case series / retrospective case series / reference laboratory series, Grade III = Exclusion of patients (i.e. most unwell patients treated) likely to significantly affect outcome / summary of case reports from literature.Diagnostic grade: Grade A = Pathogen detection through cell culture/animal inoculation. PCR positivity. IFA serological diagnosis with either sero-conversion or fourfold antibody response, Grade B = single sample with high titre/ELISA IgG/IgM, Grade C = Single high Weil Felix or IFA titre but titre non-significant for all patients / No record of significant Weil Felix or IFA titre / clinical diagnosis / unspecified.

https://doi.org/10.1371/journal.pntd.0005838.t001

Kenya were subject to less bias as they included complete prospective series of patients presenting with fever to healthcare facilities and demonstrated rates from 1.8-22.3% (Table 2).

Data from specific sub-populations are presented in Table 3. Two studies describe the importance of scrub typhus in women during pregnancy from Laos and the Thai-Myanmar border—with scrub typhus occurring in 3.6–5.4% of febrile patients [32–34]. Maternal infection with scrub typhus during pregnancy was associated with poor maternal and fetal outcomes; 2/9 (22.2%) of cases in Laos and 4/11 (36.4%) in Thailand/Myanmar suffered either abortion or stillbirth.

Among Lao patients with meningitis/encephalitis, 16.0% of those with a diagnosed bacterial cause for their infection had evidence for scrub typhus [35]. However, only 54.8% of these patients received treatment with appropriate antimicrobials during admission and the mortality rate associated with CNS complications was 13.6%. There are no data on morbidity or long-term sequelae available.

## At risk population

National surveillance data from patients in China, Japan, Korea and Taiwan suggest that the age group of 60–69 years was at highest risk of scrub typhus [18, 20, 22, 36]. In Thailand those aged 45–54 years were most commonly infected. In Japan and Thailand males were more at risk of scrub typhus but in all other countries with reports, females are more at risk. In South Korea, China, Taiwan and Thailand farmers were most at risk (38,183/54,558–70% of infections in China from 2006–2014); unfortunately such data are lacking from Japan. Age stratification in untreated mortality revealed increasing risk with increasing age, with the age classes 51–60 and >60 years old associated with a 45.6% and 59.8% mortality rate respectively [15].

First Author	Country, region	Year	Patients with scrub typhus n (%)	Total patients in study	Patient selection grade	Diagnostic Grade
Roopa [40]	India, Pondicherry	2012– 2015	225 (41.3%)	545	2	В
Borkakoty [41]	India, Arunachal Pradesh	2013	31 (96.9%)	32	3	В
Kumar [ <u>38]</u>	India, Chandigarh	2011– 2012	49 (24.4%)	201	2	A
Narvenkar [42]	India, Goa	2009– 2010	15 (34.1%)	44	2	В
Chrispal [ <u>43,</u> 44]	India, Vellore	2007– 2008	189 (47.5%)	398	2	В
Kamarasu [ <u>45]</u>	India, Tamil Nadu	2004– 2005	204 (16.1%)	1,270	2	В
Sharma [46]	India, Himachal Pradesh	2003– 2004	52 (34.7%)	150	2	В
Vaz [ <u>47]</u>	India, Jammu	2002	12 (50.0%)	24	2	В
Mueller [ <u>48]</u>	Cambodia, remote Western and Eastern provinces	2008– 2010	54 (3.7%)	1,475	2	A
Kasper [ <u>49</u> ]	Cambodia, within 50km Phnom Phenh	2006– 2009	35 (1.8%)	1,906	2	A
Mayxay [50]	Laos, northwest Laos and southern Laos	2008– 2010	122 (6.5%)	1,938	2	A
Blacksell [51]	Nepal, Kathmandu	2002– 2004	23 (22.3%)	103	2	A
Reller [52]	Sri Lanka, southern	2007	9 (1.0%)	883	2	A
Susilawati [53]	Australia, Cairns	2008– 2011	2 (0.6%)	340	2	A
Thiga [ <mark>54</mark> ]	Kenya, 6 regions	2015	76 (4.8%)	1,401	2	В

#### Table 2. Data on epidemiology of scrub typhus from fever studies (hospital based incidence).

Patient selection grade: Grade I = Population based active surveillance / community surveillance / hospital surveillance, Grade II = Prospective, consecutive patient case series with no inappropriate exclusions / non-consecutive case series / retrospective case series / reference laboratory series, Grade III = Exclusion of patients (i.e. most unwell patients treated) likely to significantly affect outcome / Summary of case reports from literature. Diagnostic grade: Grade A = Pathogen detection through cell culture/animal inoculation. PCR positivity. IFA serological diagnosis with either sero-conversion or fourfold antibody response, Grade B = single sample with high titre/ELISA IgG/IgM, Grade C = Single high Weil Felix or IFA titre but titre non-significant for all patients / No record of significant Weil Felix or IFA titre / clinical diagnosis / unspecified.

https://doi.org/10.1371/journal.pntd.0005838.t002

#### Table 3. Publications investigating scrub typhus affected specific sub-populations.

First Author	Country	Year	Population specifics	Number ST (%)	Diagnostic Grade
Dittrich <i>et al</i> . [35]	Laos, Vientiane	2003– 2011	Patients enrolled if admitted with CNS infections and lumbar puncture indicated	31/1051 (2.9%)	A
Chansamouth <i>et al.</i> [34]	Laos, Vientiane,	2006– 2010	All febrile pregnant inpatients	9/250 (3.6%)	A
McGready <i>et al.</i> [32]	Thai-Burmese border	2004– 2006	All febrile pregnant patients	11/203 (5.4%)	A
Premanatna <i>et al.</i> [29]	Sri Lanka	2008	Consecutive admission of military personnel with fever	26/49 (53%)	В
Nadjm <i>et al</i> . [ <u>55</u> ]	Vietnam, northern Vietnam	2001– 2003	Consecutive patients admitted to national referral hospital of infectious diseases, with no immediate diagnosis	251/7226 (3.5%)	В

https://doi.org/10.1371/journal.pntd.0005838.t003

# Complications and sequelae

The long-term impact of infection with scrub typhus has barely been examined. In Taiwan the hazard ratio of developing acute coronary syndrome was 1.4 (95% CI 1.1–1.8) in those with previous infection with scrub typhus compared to the general population without [37]. A recent case series from India that included patients with unexplained fever and/or multi-system involvement, found 24.4% to have scrub typhus, and 53.1% of patients with scrub typhus had acute kidney injury [38]. A retrospective cohort of severe scrub typhus cases admitted to an ICU in South India, found that respiratory complications requiring mechanical ventilation occurred in 87.9%, and that dysfunction of 3 or more organ systems occurred in 85.2% [39].

# Case fatality ratios

Case fatality ratios vary widely between countries, with those countries with easily accessible and established health systems showing lower mortality rates compared to countries with limited facilities (Fig 2 and Table 4). In a previous review, untreated scrub typhus infection was associated with an estimated mortality of 6.0% (median, range 0–70.0%) [15]. This review of treated scrub typhus, which included 39 studies and 91,692 patients found a median mortality of 1.4% (range 0–33.3%).

# Disability Adjusted Life Year (DALY) data

The burden of disease data for scrub typhus is highly limited. Only one study, from Laiwu Province in China, has calculated the DALYs associated with scrub typhus [24]. This study estimated that 13 DALYs were lost due to scrub typhus across the province (6 in males, 7 in females at a rate of 1.06/100,000). However, in this province no deaths were reported and



**Fig 2. Case fatality (%) of scrub typhus reported in published case series and studies.** This figure shows the locations from included case series and study reports in a map; the case fatality (%) reported is color coded (blue 0–1%; pale blue 1.2–6.3%; pale red 6.6–13.8%; and red 14–33.3%), and the detailed data of the studies included is summarized in Table 4. Source of map: http://www.naturalearthdata.com, accessed on the 14th July 2017. Kashmir and Arunachal Pradesh are depicted in grey with borders as red dashes, due to either disputed or indeterminate areas.

https://doi.org/10.1371/journal.pntd.0005838.g002

#### Table 4. Overview of case fatality ratios reported for treated scrub typhus.

Date of study, Author	Country	No. with ST	CFR %	Comments
2006–2012, Yang	China	308	0	DALY study in Laiwu
1995–2006, Liu	China	480	0	Fever study
2006–2014, Wu	China	54,558	0.07	National surveillance data
2012, Wei	China	29	13.8	Outbreak investigation in a Guangzhou city park
2002, Vaz	India	12	0	Outbreak investigation in soldiers
2014, Luthra	India	197	1.0	Fever study
2011, Gurung	India	63	1.6	Fever study
2008–2012, Thomas	India	262	1.9	Fever study in children
2010–2011, Mahajan	India	253	5.1	Fever study
2013, Masand	India	30	6.6	Fever study
2009–2010, Varghese	India	154	7.8	Fever study
2005–2010, Varghese	India	623	9.0	Fever study
2014, Khandelwal	India	52	9.6	Fever study in children
2007–2008, Chrispal	India	189	12.2	Fever study
2013–14, Sharma	India	228	13.6	Fever study
2011–2012, Kumar	India	49	16.3	Fever study
2014, Singh	India	13	23.1	ICU patients with pulmonary complications
2003–2004, Sharma	India	52	28.8	Outbreak investigation Himachal Pradesh
2009–2010, Narvencar	India	15	33.3	Fever study
2000, Matsui	Japan	756	0.3	National surveillance data
2007–2008, Yasunaga	Japan	210	1.0	Retrospective use of discharge data
2003–2012, Yoo	South Korea	141	0	Hospital surveillance data
1985–1990, Yi	South Korea	189	0	Clinical diagnosis only
2008–2012, Park	South Korea	30,478	0.1	National surveillance data
2004–2006, Kim	South Korea	208	0.5	Case control study at single site
2001–2011, Jang	South Korea	771	1.4	Retrospective study in single site
2000–2006, Lee	South Korea	297	6.1	Retrospective study in single site
2000–2006, Kim	South Korea	160	6.3	Scrub typhus in patients with liver disease
2003–2011, Dittrich	Laos	31	13.6	Patients with CNS symptoms
1948–1950, Bailey	Malaysia	35	0	Early trials of antibiotic use
2008, Premaratna	Sri Lanka	26	0	Outbreak investigation in soldiers
2006–2010, Wang	Taiwan	126	0	Retrospective study in single site
1952, Prezyman	Taiwan	47	0	Early trial of antibiotic use
1996, Watt	Thailand	19	0	Prospective evaluation of antimicrobial response
1985–2002, Silpapojakul	Thailand	73	1.4	Case series of children
2001–2002, Thap	Thailand	18	16.7	Patients with septic shock
2011–2012, Sriwongpan	Thailand	257	13.6	Retrospective hospital series
1965, Hazlett	Vietnam	32	0	Case series from military hospital
2001–2003, Nadjm	Vietnam	251	1.2	Hospital surveillance data

https://doi.org/10.1371/journal.pntd.0005838.t004

therefore these data cannot be extrapolated to countries such as India or Laos with evidence of scrub typhus associated mortality. A South Korean study evaluating the net benefit of a scrub typhus prevention program, estimated the cost of scrub typhus (medication and hospital costs and loss of earnings) at \$6.6 million per year in 2008 [56]. However, scrub typhus mortality in South Korea was only 0.14% and 75% of patients with a diagnosis were hospitalized. Therefore, these figures cannot be applied to other economically poorer countries where health practice is very different [17].

## Discussion

Scrub typhus represents a major cause of treatable febrile illness across Asia, but its disease incidence remains elusive. Fever remains one of the major reasons to seek healthcare in tropical regions but their causes remain ill-defined [57]. Access to updated evidence on incidence and trends for common causes of febrile illnesses is essential for guiding and informing global, regional, and national health policies. This systematic review collated all currently available literature regarding the disease burden and economic impact of scrub typhus and the result is sobering; there are very few studies and they have great heterogeneity in methodology.

Acquisition of estimates for incidence and mortality proved difficult, as numerators had varying levels of confidence in diagnosis or denominators were either absent, or required further extrapolation. Ideally, data derived from population-based surveillance studies would be graded considerably higher than from hospital-based surveillance, but unfortunately no non-hospital-based surveillance data are publicly available for scrub typhus—unlike for diseases like typhoid where these data are readily available for various countries [58]. Further, the epidemiology of scrub typhus within a country is heterogeneous—the pronounced seasonality of these diseases and the changing urban/rural distribution, with defined areas of high infected mite intensities (mite islands) challenge the common approaches of disease incidence evaluation [59, 60]. Febrile illness surveillance should be performed in multiple representative areas, ideally covering one full calendar year before inferences on national disease incidence can be made [58].

## Incidence surveillance

Only 5 countries have established scrub typhus surveillance systems. All of these have shown an increasing minimum incidence of scrub typhus over recent years, with increasing evidence of shift towards urbanized areas. However, the apparent increase in minimum incidence is confounded by local enhanced knowledge of the disease and it remains uncertain whether these data reflect true de novo emerging disease or emerging awareness of a pre-existing disease. Surveillance systems also use diverse diagnostic tests and therefore inter-country comparisons are not always possible. There are no data on whether these surveillance systems have been evaluated to determine an estimate of missed cases, however it is likely that the numbers are conservative estimates. Regardless of these flaws, surveillance systems are an essential part of disease control strategies. Improved febrile disease surveillance providing national data should be initiated in more afflicted countries, as this would result in morbidity and mortality data that could be used to direct healthcare resources, future vaccine demand and delivery and assessment of effectiveness of any control programs.

Clearly, striving towards improved surveillance should be key, with a focus on providing reliable numerators (using diagnostic assays with suitable sensitivities and specificities), and representative denominators (well-defined target populations). Additionally, no 'multiplier data' or 'multiplier studies' are available—these are considered to improve estimation of incidence by using healthcare utilization surveys and to correct for under-ascertainment in healthcare facility studies [58].

## Seroprevalence

Seroprevalence data was available from 5 countries only–indicating high background exposure levels, and therefore a high probability that larger numbers of unidentified and/or asymptomatic infections occur. Disease seroprevalence data must be interpreted with caution due to unknown antibody dynamics over time and uncertainty as to whether those seropositive became sick or were asymptomatic. In scrub typhus, both humoral and cell-mediated protective immune responses wane over time, but detailed understanding of this remains elusive [61]. Moreover, the population-wide frequencies of patients with reversion to seronegativity and potential disease susceptibility remain unknown, and therefore the actual exposure in these studies is likely to be substantially higher [62].

## **Fever studies**

Scrub typhus is a leading cause of treatable non-malarial febrile illness in prospective fever etiology studies (n = 14). An increasing number of studies have unraveled the major contribution of scrub typhus to the febrile illness burden. However, the large variation of scrub typhus rates in prospective fever studies (median 23.4% IQR 5.2–39.7 ranging from 1–96.9% depending on country and patient selection), reflect a lack of standardization and comparability among study designs and diagnostic modalities used. None of these studies have used modeling or extrapolation to take into account data from healthcare utilization surveys, which may give a more accurate idea of numbers of people with scrub typhus. In addition, recent studies have raised concern on the persistence of *O. tsutsugamushi* after treatment, especially using bacteriostatic drugs such as tetracyclines and macrolides [63, 64].

## Vulnerable populations and case fatality ratios

Based on very limited data, scrub typhus is likely to have considerable impact on vulnerable populations-the median untreated mortality of scrub typhus in the elderly was ~29%—approximately 5-fold higher compared to the overall population mortality of 6% [15]. In women with scrub typhus during pregnancy, miscarriages occurred in 17% and poor neonatal outcomes in 42% of cases, which is more severe than the consequences of malaria in pregnancy [65]. Further, the mortality in patients requiring a lumbar puncture for scrub typhus CNS complications in Laos was 14% [59]. Scrub typhus is usually an easily treatable disease and the majority of these complications could be prevented by early recognition/diagnosis and increased usage of empirical doxycycline [66].

It is difficult to draw any definitive conclusions from the case-fatality data due to the heterogeneity in studies. They range from national surveillance data to case series of those admitted to ICU. National surveillance data from China, Japan and Korea provide *case fatality ratios* of 0.068–0.26%. However, the health facilities in these countries are significantly more advanced than other endemic countries. The fever studies from South India provide estimates of case fatality risk, but they vary from 0–33.3%—importantly, these data included patients who presented to hospital and therefore will miss those that do not have severe disease.

## Health economics

DALY data are lacking in all countries except from one area of China, where a rate of 1.06/ 100,000 people was found, with a zero mortality rate. Case series and studies from Taiwan and India examining long-term complications, imply that the mortality and morbidity from scrub typhus is under-recognized and that possible long term consequences may occur many years later, and may be important contributors to the overall DALY burden [37, 67]. Despite scrub typhus being the foremost cause of treatable febrile illness in Asia it is not evaluated by the Global Burden of Disease studies [68].

## Study limitations

This study involved an extensive search of the literature and includes up-to-date and relevant studies. However, there are several limitations; as English is not the native language in the

majority of countries where scrub typhus is endemic, there is a potential bulk of relevant literature that is not indexed in the databases used. The risks of publication bias and the heterogeneity of methods and reporting in the articles limit the conclusions. Specific difficulties relating to the diagnosis of scrub typhus suggest that studies reporting data from national surveillance systems are likely to suffer from missing data due to those that do not seek medical attention are misdiagnosed or not reported. The majority of fever studies suffers from selection bias and often relies on suboptimal diagnostic tools.

## Future focus

Reports from Africa, the Middle East and most recently South America, suggest that scrub typhus is more widespread than previously appreciated. The molecular detection of *Orientia* spp. in rodents from Southern France and Senegal suggest that rodent-mite cycles could maintain the pathogen in nature but whether these *Orientia* spp. represent human pathogens is unknown [69]. The countries most affected by scrub typhus are currently experiencing profound demographic, economic and ecological changes [70]. Deforestation, growing cities and climate change may lead to migration of rodents carrying infected mites and expand to more urban and non-endemic areas [8, 11, 16]. Recently the impact of an earthquake on exposing the population to the possibly perturbed soil dwelling vectors causing scrub typhus was highlighted in Nepal [71].

Ancestor *et al.* mapped non-malarial causes of fever, including scrub typhus, in the Mekong region [2]. Kelly *et al.* developed a vector map of scrub typhus based on literature review to include probable and confirmed cases that included geo-referenced locations [72]. These are useful resources that can be built upon to estimate incidence in areas where data is limited. In scrub typhus the extracted information of studies from the 1940s requires careful consideration to identify what data are clinically relevant today. Derne *et al.* summarized and mapped the distribution of rickettsia and their vectors in Oceania, confirming the widespread presence and providing a scaffold to build upon [73]. Ideally, concerted efforts in providing well maintained up-to-date mapping of human cases and vector (chigger mite) distribution would contribute substantially to understanding the burden of disease.

Burden of disease studies often use syndromic 'envelopes' for certain conditions (for example "diarrhea" or "fever"). Developing a fever 'envelope' approach for estimating its burden of disease, in conjunction with detailed fever etiology studies would provide improved, standardized and globally comparable incidence data [74, 75]. The resulting data could be stratified further and would inform on the actual burden of disease, as well as provide valuable baseline data to support economic evaluations and mathematical modeling of future interventions [76]. For example, an incentive for identifying endemic areas of scrub typhus may result in increasing cost-effectiveness of rapid diagnostic test (RDT) use. Testing for frequent bacterial pathogens is likely to be economical, reducing hospitalization rates, and informs not only treatment requirements, but also appropriate antibiotic usage [77].

In the case of dengue, the quality of data available has improved substantially and in 2010 there were an estimated 96 million apparent and 294 million unapparent dengue infections globally [78]. Although dengue and scrub typhus both top the list of fever etiologies in multiple studies in Asia, the more easily-treatable disease is neglected–it is time for more integrated expert collaborative research to provide these urgently needed objective data [57, 78, 79].

These data-despite their limitations-make a case for scrub typhus as an important neglected tropical disease of mainly rural populations, with an increasing urban proportion. In countries with established surveillance systems, the reported incidence is increasing and robust documentation of scrub typhus in Chile suggests a much wider global presence than previously understood.

The lack of data on global incidence and disease burden highlights the need for this treatable infection to receive increased attention and research to inform health policy.

## **Supporting information**

**S1 File. PRISMA checklist.** (DOC)

## Acknowledgments

We are grateful to Dr. Tri Wangrangsimakul for his kind support in acquiring and translating Thai epidemiological information on scrub typhus.

## **Author Contributions**

Conceptualization: Yoel Lubell, Daniel H. Paris.

Data curation: Ana Bonell, Paul N. Newton, John A. Crump, Daniel H. Paris.

Formal analysis: Ana Bonell, Yoel Lubell, Paul N. Newton, John A. Crump, Daniel H. Paris.

Investigation: Ana Bonell.

Methodology: Ana Bonell, Daniel H. Paris.

Software: Ana Bonell, Daniel H. Paris.

Supervision: Daniel H. Paris.

Visualization: Daniel H. Paris.

Writing - original draft: Ana Bonell, Daniel H. Paris.

Writing – review & editing: Ana Bonell, Yoel Lubell, Paul N. Newton, John A. Crump, Daniel H. Paris.

## References

- 1. Paris DH, Day NPJ. 22—Tropical Rickettsial Infections. In: White JFJHJKLJ, editor. Manson's Tropical Infectious Diseases (Twenty-Third Edition). London: W.B. Saunders; 2014. p. 273–91.e5.
- Acestor N, Cooksey R, Newton PN, Menard D, Guerin PJ, Nakagawa J, et al. Mapping the aetiology of non-malarial febrile illness in Southeast Asia through a systematic review—terra incognita impairing treatment policies. PLoS One. 2012; 7(9):6.
- 3. Kawamura A Jr., Tanaka H, Tamura A. Tsutsugamushi Disease. University of Tokyo Press. 1995.
- Izzard L, Fuller A, Blacksell SD, Paris DH, Richards AL, Aukkanit N, et al. Isolation of a novel Orientia species (O. chuto sp. nov.) from a patient infected in Dubai. Journal of Clinical Microbiology. 2010; 48 (12):4404–9. Epub 2010/10/12. https://doi.org/10.1128/JCM.01526-10 PMID: 20926708; PubMed Central PMCID: PMC3008486.
- Balcells ME, Rabagliati R, García P, Poggi H, Oddó D, Concha M, et al. Endemic Scrub Typhus-like Illness, Chile. Emerg Infect Dis. 2011; 17(9):1659–63. https://doi.org/10.3201/eid1709.100960 PMID: 21888791.
- Cosson JF, Galan M, Bard E, Razzauti M, Bernard M, Morand S, et al. Detection of Orientia sp. DNA in rodents from Asia, West Africa and Europe. Parasit Vectors. 2015; 8:172. https://doi.org/10.1186/ s13071-015-0784-7 PMID: 25884521; PubMed Central PMCID: PMC4374543.
- Maina AN, Farris CM, Odhiambo A, Jiang J, Laktabai J, Armstrong J, et al. Q Fever, Scrub Typhus, and Rickettsial Diseases in Children, Kenya, 2011–2012. Emerg Infect Dis. 2016; 22(5):883–6. https://doi. org/10.3201/eid2205.150953 PMID: 27088502; PubMed Central PMCID: PMCPMC4861507.
- Walker DH. Scrub Typhus—Scientific Neglect, Ever-Widening Impact. New England Journal of Medicine. 2016; 375(10):913–5. https://doi.org/10.1056/NEJMp1608499 PMID: 27602663.

- Noh M, Lee Y, Chu C, Gwack J, Youn SK, Huh S. Are there spatial and temporal correlations in the incidence distribution of scrub typhus in Korea? Osong Public Health Res Perspect. 2013; 4(1):39–44. https://doi.org/10.1016/j.phrp.2013.01.002 PMID: 24159528; PubMed Central PMCID: PMCPMC3747676.
- Wei Y, Huang Y, Luo L, Xiao X, Liu L, Yang Z. Rapid increase of scrub typhus: an epidemiology and spatial-temporal cluster analysis in Guangzhou City, Southern China, 2006–2012. PLoS One. 2014; 9 (7):e101976. https://doi.org/10.1371/journal.pone.0101976 PMID: 25006820; PubMed Central PMCID: PMCPMC4090214.
- Yang LP, Liu J, Wang XJ, Ma W, Jia CX, Jiang BF. Effects of meteorological factors on scrub typhus in a temperate region of China. Epidemiol Infect. 2014; 142(10):2217–26. Epub 2014/05/08. https://doi. org/10.1017/S0950268813003208 PMID: 24800904.
- Isaac R, Varghese GM, Mathai E, J M, Joseph I. Scrub typhus: prevalence and diagnostic issues in rural Southern India. Clin Infect Dis. 2004; 39(9):1395–6. https://doi.org/10.1086/424748 PMID: 15494919.
- World Health Organization Department of communicable disease surveillance and response. WHO Recommended Surveillance Standards.vol 2. WHO/CDS/CSR/ISR/99.2. 1999.
- Rosenberg R. Drug-resistant scrub typhus: Paradigm and paradox. Parasitol Today. 1997; 13(4):131– 2. PMID: 15275098.
- Taylor AJ, Paris DH, Newton PN. A Systematic Review of Mortality from Untreated Scrub Typhus (*Orientia tsutsugamushi*). PLoS Negl Trop Dis. 2015; 9(8):e0003971. https://doi.org/10.1371/journal. pntd.0003971 PMID: 26274584; PubMed Central PMCID: PMCPMC4537241.
- Park SW, Ha NY, Ryu B, Bang JH, Song H, Kim Y, et al. Urbanization of scrub typhus disease in South Korea. PLoS Negl Trop Dis. 2015; 9(5):e0003814. Epub 2015/05/23. https://doi.org/10.1371/journal. pntd.0003814 PMID: 26000454; PubMed Central PMCID: PMCPmc4441427.
- Park JH, Kim SJ, Youn SK, Park K, Gwack J. Epidemiology of scrub typhus and the eschars patterns in South Korea from 2008 to 2012. Jpn J Infect Dis. 2014; 67(6):458–63. PMID: 25410561.
- Lee HW, Cho PY, Moon SU, Na BK, Kang YJ, Sohn Y, et al. Current situation of scrub typhus in South Korea from 2001–2013. Parasites & vectors. 2015; 8:238. Epub 2015/05/01. https://doi.org/10.1186/ s13071-015-0858-6 PMID: 25928653; PubMed Central PMCID: PMCPmc4416255.
- Matsui T, Kramer MH, Mendlein JM, Osaka K, Ohyama T, Takahashi H, et al. Evaluation of National Tsutsugamushi Disease Surveillance—Japan, 2000. Jpn J Infect Dis. 2002; 55(6):197–203. Epub 2003/02/28. PMID: 12606829.
- Yasunaga H, Horiguchi H, Kuwabara K, Hashimoto H, Matsuda S. Delay in tetracycline treatment increases the risk of complications in Tsutsugamushi disease: data from the Japanese Diagnosis Procedure Combination database. Intern Med. 2011; 50(1):37–42. PMID: 21212571.
- Bureau of Epidemiology, DDC, MPH [cited 2016 22nd Oct]. Available from: <a href="http://www.boe.moph.go.th/boedb/surdata/disease.php?dcontent=situation&ds=44">http://www.boe.moph.go.th/boedb/surdata/disease.php?dcontent=situation&ds=44</a>.
- 22. Wu YC, Qian Q, Magalhaes RJ, Han ZH, Haque U, Weppelmann TA, et al. Rapid Increase in Scrub Typhus Incidence in Mainland China, 2006–2014. Am J Trop Med Hyg. 2016; 94(3):532–6. https://doi. org/10.4269/ajtmh.15-0663 PMID: 26711517; PubMed Central PMCID: PMCPMC4775887.
- Lyu Y, Tian L, Zhang L, Dou X, Wang X, Li W, et al. A case-control study of risk factors associated with scrub typhus infection in Beijing, China. PLoS One. 2013; 8(5):e63668. https://doi.org/10.1371/journal. pone.0063668 PMID: 23691083; PubMed Central PMCID: PMC3653850.
- Yang LP, Liang SY, Wang XJ, Li XJ, Wu YL, Ma W. Burden of disease measured by disability-adjusted life years and a disease forecasting time series model of scrub typhus in Laiwu, China. PLoS Negl Trop Dis. 2015; 9(1):e3420. Epub 2015/01/09. https://doi.org/10.1371/journal.pntd.0003420 PMID: 25569248; PubMed Central PMCID: PMCPmc4288724.
- De W, Jing K, Huan Z, Qiong ZH, Monagin C, Min ZJ, et al. Scrub Typhus, a Disease with Increasing Threat in Guangdong, China. PLoS ONE. 2015; 10(2):e0113968. doi: <u>10.1371/journal.pone.0113968</u>. PMC4331496. PMID: 25689778
- Maude RR, Maude RJ, Ghose A, Amin MR, Islam MB, Ali M, et al. Serosurveillance of Orientia tsutsugamushi and Rickettsia typhi in Bangladesh. Am J Trop Med Hyg. 2014; 91(3):580–3. https://doi.org/ 10.4269/ajtmh.13-0570 PMID: 25092819; PubMed Central PMCID: PMCPMC4155564.
- 27. Vallee J, Thaojaikong T, Moore CE, Phetsouvanh R, Richards AL, Souris M, et al. Contrasting spatial distribution and risk factors for past infection with scrub typhus and murine typhus in Vientiane City, Lao PDR. PLoS Negl Trop Dis. 2010; 4(12):e909. Epub 2010/12/15. https://doi.org/10.1371/journal.pntd. 0000909 PMID: 21151880; PubMed Central PMCID: PMCPmc2998433.
- 28. Tay ST, Mohamed Zan HA, Lim YAL, Ngui R. Antibody Prevalence and Factors Associated with Exposure to *Orientia tsutsugamushi* in Different Aboriginal Subgroups in West Malaysia. PLoS Neglected

Tropical Diseases. 2013; 7(8):e2341. doi: 10.1371/journal.pntd.0002341. PMC3731234. PMID: 23936576

- Premaratna R, Ariyaratna N, Attanayake C, Bandara W, Chandrasena N, de Silva HJ. Rickettsial infection among military personnel deployed in Northern Sri Lanka. BMC Infect Dis. 2014; 14:3864. <a href="https://doi.org/10.1186/s12879-014-0688-8">https://doi.org/10.1186/s12879-014-0688-8</a> PMID: 25527099; PubMed Central PMCID: PMCPMC4335424.
- Richards AL, Ratiwayanto S, Rahardjo E, Kelly DJ, Dasch GA, Fryauff DJ, et al. Serologic evidence of infection with ehrlichiae and spotted fever group rickettsiae among residents of Gag Island, Indonesia. Am J Trop Med Hyg. 2003; 68(4):480–4. PMID: 12875301.
- Spicer PE, Taufa T, Benjamin AL. Scrub typhus (*Orientia tsutsugamushi*), spotted fever (Rickettsia australis) and dengue fever as possible causes of mysterious deaths in the Strickland Gorge area of Southern Highlands and West Sepik Provinces of Papua New Guinea. P N G Med J. 2007; 50(3–4):172–83. PMID: 19583101.
- McGready R, Ashley EA, Wuthiekanun V, Tan SO, Pimanpanarak M, Viladpai-Nguen SJ, et al. Arthropod borne disease: the leading cause of fever in pregnancy on the Thai-Burmese border. PLoS Negl Trop Dis. 2010; 4(11):0000888.
- 33. McGready R, Prakash JA, Benjamin SJ, Watthanaworawit W, Anantatat T, Tanganuchitcharnchai A, et al. Pregnancy outcome in relation to treatment of murine typhus and scrub typhus infection: a Fever cohort and a case series analysis. PLoS Negl Trop Dis. 2014; 8(11).
- Chansamouth V, Thammasack S, Phetsouvanh R, Keoluangkot V, Moore CE, Blacksell SD, et al. The Aetiologies and Impact of Fever in Pregnant Inpatients in Vientiane, Laos. PLoS Negl Trop Dis. 2016; 10(4):e0004577. https://doi.org/10.1371/journal.pntd.0004577 PMID: 27050192; PubMed Central PMCID: PMCPMC4822858.
- 35. Dittrich S, Rattanavong S, Lee SJ, Panyanivong P, Craig SB, Tulsiani SM, et al. Orientia, rickettsia, and leptospira pathogens as causes of CNS infections in Laos: a prospective study. The Lancet Global Health. 3(2):e104–e12. https://doi.org/10.1016/S2214-109X(14)70289-X PMID: 25617190
- Lee YS, Wang PH, Tseng SJ, Ko CF, Teng HJ. Epidemiology of scrub typhus in eastern Taiwan, 2000– 2004. Jpn J Infect Dis. 2006; 59(4):235–8. PMID: 16936341.
- Chung WS, Lin CL, Hsu WH, Kao CH. Scrub typhus increases the risk of developing acute coronary syndrome: a nationwide cohort study. Heart. 2014; 100(23):1844–50. <u>https://doi.org/10.1136/heartjnl-</u> 2014-306181 PMID: 25147287.
- Kumar V, Kumar V, Yadav AK, Iyengar S, Bhalla A, Sharma N, et al. Scrub typhus is an under-recognized cause of acute febrile illness with acute kidney injury in India. PLoS Negl Trop Dis. 2014; 8(1): e2605. Epub 2014/02/06. https://doi.org/10.1371/journal.pntd.0002605 PMID: 24498445; PubMed Central PMCID: PMCPmc3907360.
- Griffith M, Peter JV, Karthik G, Ramakrishna K, Prakash JAJ, Kalki RC, et al. Profile of organ dysfunction and predictors of mortality in severe scrub typhus infection requiring intensive care admission. Indian J Crit Care Med. 2014; 18(8):497–502. https://doi.org/10.4103/0972-5229.138145 PMID: 25136187; PubMed Central PMCID: PMCPMC4134622.
- Roopa KS, Karthika K, Sugumar M, Bammigatti C, Shamanna SB, Harish BN. Serodiagnosis of Scrub Typhus at a Tertiary Care Hospital from Southern India. Journal of Clinical and Diagnostic Research: JCDR. 2015; 9(11):DC05–DC7. doi: <u>10.7860/JCDR/2015/15871.6759</u>. PMC4668406. PMID: 26675629
- **41.** Borkakoty B, Jakharia A, Biswas D, Mahanta J. Co-infection of scrub typhus and leptospirosis in patients with pyrexia of unknown origin in Longding district of Arunachal Pradesh in 2013. Indian J Med Microbiol. 2016; 34(1):88–91. https://doi.org/10.4103/0255-0857.174116 PMID: 26776126.
- Narvencar KP, Rodrigues S, Nevrekar RP, Dias L, Dias A, Vaz M, et al. Scrub typhus in patients reporting with acute febrile illness at a tertiary health care institution in Goa. Indian J Med Res. 2012; 136 (6):1020–4. PMID: 23391799; PubMed Central PMCID: PMCPMC3612306.
- 43. Chrispal A, Boorugu H, Gopinath KG, Chandy S, Prakash JA, Thomas EM, et al. Acute undifferentiated febrile illness in adult hospitalized patients: the disease spectrum and diagnostic predictors—an experience from a tertiary care hospital in South India. Trop Doct. 2010; 40(4):230–4. <u>https://doi.org/10.1258/td.2010.100132 PMID: 20870680</u>.
- Chrispal A, Boorugu H, Gopinath KG, Prakash JA, Chandy S, Abraham OC, et al. Scrub typhus: an unrecognized threat in South India—clinical profile and predictors of mortality. Trop Doct. 2010; 40 (3):129–33. https://doi.org/10.1258/td.2010.090452 PMID: 20360426.
- Kamarasu K, Malathi M, Rajagopal V, Subramani K, Jagadeeshramasamy D, Mathai E. Serological evidence for wide distribution of spotted fevers & typhus fever in Tamil Nadu. Indian J Med Res. 2007; 126 (2):128–30. PMID: <u>17932437</u>.

- Sharma A, Mahajan S, Gupta ML, Kanga A, Sharma V. Investigation of an outbreak of scrub typhus in the himalayan region of India. Jpn J Infect Dis. 2005; 58(4):208–10. Epub 2005/08/24. PMID: 16116251.
- Vaz LS, Gupta NK. Outbreak of Scrub Typhus in Jammu A Report. Medical Journal Armed Forces India. 62(4):342–3. https://doi.org/10.1016/S0377-1237(06)80103-6
- Mueller TC, Siv S, Khim N, Kim S, Fleischmann E, Ariey F, et al. Acute undifferentiated febrile illness in rural Cambodia: a 3-year prospective observational study. PLoS One. 2014; 9(4):e95868. https://doi. org/10.1371/journal.pone.0095868 PMID: 24755844; PubMed Central PMCID: PMCPMC3995936.
- Kasper MR, Blair PJ, Touch S, Sokhal B, Yasuda CY, Williams M, et al. Infectious etiologies of acute febrile illness among patients seeking health care in south-central Cambodia. Am J Trop Med Hyg. 2012; 86(2):246–53. Epub 2012/02/04. https://doi.org/10.4269/ajtmh.2012.11-0409 PMID: 22302857; PubMed Central PMCID: PMCPmc3269275.
- Mayxay M, Castonguay-Vanier J, Chansamouth V, Dubot-Peres A, Paris DH, Phetsouvanh R, et al. Causes of non-malarial fever in Laos: a prospective study. Lancet Glob Health. 2013; 1(3):e46–e54.
- Blacksell SD, Sharma NP, Phumratanaprapin W, Jenjaroen K, Peacock SJ, White NJ, et al. Serological and blood culture investigations of Nepalese fever patients. Trans R Soc Trop Med Hyg. 2007; 101 (7):686–90. https://doi.org/10.1016/j.trstmh.2007.02.015 PMID: 17433390.
- Reller ME, Bodinayake C, Nagahawatte A, Devasiri V, Kodikara-Arachichi W, Strouse JJ, et al. Unsuspected rickettsioses among patients with acute febrile illness, Sri Lanka, 2007. Emerg Infect Dis. 2012; 18(5):825–9. https://doi.org/10.3201/eid1805.111563 PMID: 22516455; PubMed Central PMCID: PMCPMC3358078.
- Susilawati TN, McBride WJH. Undiagnosed undifferentiated fever in Far North Queensland, Australia: a retrospective study. International Journal of Infectious Diseases. 2014; 27:59–64. <u>https://doi.org/10.1016/j.ijid.2014.05.022</u> PMID: 25173425
- 54. Thiga JW, Mutai BK, Eyako WK, Ng'ang'a Z, Jiang J, Richards AL, et al. High seroprevalence of antibodies against spotted fever and scrub typhus bacteria in patients with febrile Illness, Kenya. Emerging infectious diseases. 2015; 21(4):688–91. https://doi.org/10.3201/eid2104.141387 PMID: 25811219
- 55. Nadjm B, Thuy PT, Trang VD, Ha le D, Kinh NV, Wertheim HF. Scrub typhus in the northern provinces of Vietnam: an observational study of admissions to a national referral hospital. Trans R Soc Trop Med Hyg. 2014; 108(11):739–40. <u>https://doi.org/10.1093/trstmh/tru145</u> PMID: <u>25253616</u>; PubMed Central PMCID: PMCPMC4191438.
- 56. Kim J, Lee E, Rhee HC. Cost-benefit analysis of the tsutsugamushi disease prevention program in South Korea. Jpn J Infect Dis. 2012; 65(3):222–7. PMID: 22627303.
- Crump JA. Time for a comprehensive approach to the syndrome of fever in the tropics. Trans R Soc Trop Med Hyg. 2014; 108(2):61–2. https://doi.org/10.1093/trstmh/trt120 PMID: 24463580; PubMed Central PMCID: PMCPMC3916746.
- Crump JA, Youssef FG, Luby SP, Wasfy MO, Rangel JM, Taalat M, et al. Estimating the incidence of typhoid fever and other febrile illnesses in developing countries. Emerg Infect Dis. 2003; 9(5):539–44. https://doi.org/10.3201/eid0905.020428 PMID: 12737736; PubMed Central PMCID: PMCPMC2972755.
- 59. Dittrich S, Rattanavong S, Lee SJ, Panyanivong P, Craig SB, Tulsiani SM, et al. Orientia, rickettsia, and leptospira pathogens as causes of CNS infections in Laos: a prospective study. Lancet Glob Health. 2015; 3(2):e104–12. https://doi.org/10.1016/S2214-109X(14)70289-X PMID: 25617190; PubMed Central PMCID: PMCPMC4547322.
- 60. Vallee J, Thaojaikong T, Moore CE, Phetsouvanh R, Richards AL, Souris M, et al. Contrasting spatial distribution and risk factors for past infection with scrub typhus and murine typhus in Vientiane City, Lao PDR. PLoS Negl Trop Dis. 2010; 4(12):e909. Epub 2010/12/15. https://doi.org/10.1371/journal.pntd. 0000909 PMID: 21151880; PubMed Central PMCID: PMC2998433.
- Cho B-A, Ko Y, Kim Y-S, Kim S, Choi M-S, Kim I-S, et al. Phenotypic Characterization of Peripheral T Cells and Their Dynamics in Scrub Typhus Patients. PLoS Neglected Tropical Diseases. 2012; 6(8): e1789. doi: 10.1371/journal.pntd.0001789. PMC3419201. PMID: 22905277
- Bourgeois AL, Olson JG, Fang RC, Huang J, Wang CL, Chow L, et al. Humoral and cellular responses in scrub typhus patients reflecting primary infection and reinfection with *Rickettsia tsutsugamushi*. Am J Trop Med Hyg. 1982; 31(3 Pt 1):532–40. PMID: 6805348.
- Chung M-H, Lee J-S, Baek J-H, Kim M, Kang J-S. Persistence of Orientia tsutsugamushi in Humans. J Korean Med Sci. 2012; 27(3):231–5. https://doi.org/10.3346/jkms.2012.27.3.231 PMID: 22379331; PubMed Central PMCID: PMCPMC3286767.
- Im J-H, Baek JH, Lee J-S, Chung M-H, Lee SM, Kang J- S. In Vitro Bacteriostatic Effects of Rifampin on Orientia tsutsugamushi. J Korean Med Sci. 2014; 29(2):183–9. https://doi.org/10.3346/jkms.2014.29.2. 183 PMID: 24550643; PubMed Central PMCID: PMCPMC3923995.

- 65. McGready R, Prakash JA, Benjamin SJ, Watthanaworawit W, Anantatat T, Tanganuchitcharnchai A, et al. Pregnancy outcome in relation to treatment of murine typhus and scrub typhus infection: a fever cohort and a case series analysis. PLoS Negl Trop Dis. 2014; 8(11):e3327. https://doi.org/10.1371/journal.pntd.0003327 PMID: 25412503; PubMed Central PMCID: PMCPMC4238995.
- 66. Cross R, Ling C, Day NP, McGready R, Paris DH. Revisiting doxycycline in pregnancy and early child-hood—time to rebuild its reputation? Expert Opin Drug Saf. 2016; 15(3):367–82. https://doi.org/10.1517/14740338.2016.1133584 PMID: 26680308; PubMed Central PMCID: PMCPMC4898140.
- Dogra S. Recent advances in understanding pathophysiology of scrub typhus. JK Science. 2010; 12 (2):70–1.
- Murray CJ, Vos T, Lozano R, Naghavi M, Flaxman AD, Michaud C, et al. Disability-adjusted life years (DALYs) for 291 diseases and injuries in 21 regions, 1990–2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012; 380(9859):2197–223. Epub 2012/12/19. <u>https://doi.org/ 10.1016/S0140-6736(12)61689-4 PMID: 23245608</u>.
- Cosson JF, Galan M, Bard E, Razzauti M, Bernard M, Morand S, et al. Detection of Orientia sp. DNA in rodents from Asia, West Africa and Europe. Parasit Vectors. 2015; 8:172. <u>https://doi.org/10.1186/</u> s13071-015-0784-7 PMID: 25884521; PubMed Central PMCID: PMCPMC4374543.
- 70. Tsai PJ, Yeh HC. Scrub typhus islands in the Taiwan area and the association between scrub typhus disease and forest land use and farmer population density: geographically weighted regression. BMC Infect Dis. 2013; 13:191. <u>https://doi.org/10.1186/1471-2334-13-191</u> PMID: 23627966; PubMed Central PMCID: PMC3648375.
- Basnyat B. Typhoid versus typhus fever in post-earthquake Nepal. Lancet Glob Health. 2016; 4(8): e516–7. https://doi.org/10.1016/S2214-109X(16)30094-8 PMID: 27443773.
- 72. Kelly DJ, Fuerst PA, Ching WM, Richards AL. Scrub typhus: the geographic distribution of phenotypic and genotypic variants of *Orientia tsutsugamushi*. Clin Infect Dis. 2009; 48 Suppl 3:S203–30. https:// doi.org/10.1086/596576 PMID: 19220144.
- 73. Derne B, Weinstein P, Musso D, Lau C. Distribution of rickettsioses in Oceania: past patterns and implications for the future. Acta Trop. 2015; 143:121–33. <u>https://doi.org/10.1016/j.actatropica.2014.10.012</u> PMID: 25446172.
- Crump JA, Kirk MD. Estimating the Burden of Febrile Illnesses. PLoS Negl Trop Dis. 2015; 9(12): e0004040. https://doi.org/10.1371/journal.pntd.0004040 PMID: 26633014; PubMed Central PMCID: PMCPMC4668833.
- 75. Kretzschmar M, Mangen MJ, Pinheiro P, Jahn B, Fevre EM, Longhi S, et al. New methodology for estimating the burden of infectious diseases in Europe. PLoS Med. 2012; 9(4):e1001205. https://doi.org/ 10.1371/journal.pmed.1001205 PMID: 22529750; PubMed Central PMCID: PMCPMC3328443.
- 76. Van Kerkhove MD, Asikainen T, Becker NG, Bjorge S, Desenclos J- C, dos Santos T, et al. Studies needed to address public health challenges of the 2009 H1N1 influenza pandemic: insights from modeling. PLoS Med. 2010; 7(6):e1000275. https://doi.org/10.1371/journal.pmed.1000275 PMID: 20532237; PubMed Central PMCID: PMCPMC2879409.
- 77. Lubell Y, Althaus T, Blacksell SD, Paris DH, Mayxay M, Pan-Ngum W, et al. Modelling the Impact and Cost-Effectiveness of Biomarker Tests as Compared with Pathogen-Specific Diagnostics in the Management of Undifferentiated Fever in Remote Tropical Settings. PLoS One. 2016; 11(3):e0152420. https://doi.org/10.1371/journal.pone.0152420 PMID: 27027303; PubMed Central PMCID: PMCPMC4814092.
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, Moyes CL, et al. The global distribution and burden of dengue. Nature. 2013; 496(7446):504–7. https://doi.org/10.1038/nature12060 PMID: 23563266; PubMed Central PMCID: PMCPMC3651993.
- 79. Tarantola A, Goutard F, Newton P, de Lamballerie X, Lortholary O, Cappelle J, et al. Estimating the burden of Japanese encephalitis virus and other encephalitides in countries of the mekong region. PLoS Negl Trop Dis. 2014; 8(1):e2533. https://doi.org/10.1371/journal.pntd.0002533 PMID: 24498443.